Author’s response to reviews

Title: Factors Associated with Mortality Risk for Malignant Colonic Obstruction in Elderly Patients

Authors:

  guo minggao (guominggao203@hotmail.com)
  Liu Jia Zhe (zhz45@sina.com)
  Zheng Qi (zheng1863@sina.com)
  Di Jian Zhong (dijianzhong@163.com)
  Wang Yu (wanyu132@hotmail.com)
  Fan You Ben (fanyouben2006@163.com)
  Huang Xin Yu (huangxy11@hotmail.com)
  Fen Yi (orchidfy@hotmail.com)

Version: 5 Date: 10 April 2014

Author’s response to reviews: see over
Shanghai, Apr. 10, 2014

Re: MS: 1338564953113264

Dear Professor Tonilynn Manibo:

Thank you for commenting our manuscript (MS: 1338564953113264, entitled “Factors Predicting the Mortality Risk for Malignant Colonic Obstruction in Elderly Patients”). The comments were constructive and helped us improve our work. And we have improved this manuscript accordingly. For the detailed responses please see the following point-by-point description of all changes.

Sincerely,

Guo Minggao

Comments: Please use initials only to refer to authors in Authors Contribution section, e.g. Fen Yi : FY.

ANSWER: According to the Executive Editor’s comment, we revised it in the new manuscript (page 12).

Comments: Please also ensure that your revised manuscript conforms to the journal style.

ANSWER: According to the Executive Editor’s comment, we revised it in Abstract (page 2,3) and Conclusions (page 12) in the new manuscript.